Author's response to reviews

Title: Cingulate cortex hypoperfusion predicts Alzheimer's disease in mild cognitive impairment

Authors:

Chaorui Huang (chaorui.huang@neurotec.ki.se)
Lars-Olof Wahlund (lars-olof.wahlund@neurotec.ki.se)
Leif Svensson (Leif@asf.hs.sll.se)
Bengt Winblad (bengt.winblad@neurotec.ki.se)
Per Julin (per.julin@neurotec.ki.se)

Version: 2 Date: 6 Sep 2002

#1. Remove the co-authors qualifications from the title paper.
#2. The reference numbers are placed before the full stop, at the end of the sentence.
#3. Introduction was changed to Background.
#4. Author contributorship' was added following the 'Competing interests'.
#5. The two figures were cropped.
#6. Figure legends were pasted after the 'References'.
#7. The more detailed informations of subject recruitment, inclusion and exclusion criteria were added under the 'Subject selection', page 4, paragraph 3.
#8. The conversion rate was added in the 'Results' section, page 7, line 3, and 'Discussion' section, page 7, last paragraph.
#9. The informations of psychotropic and cholinesterase inhibitor medication were described under the 'Subject selection', page 4, paragraph 3.
#10. We did not find correlations between MMSE and rCBF, so we would prefer not to include the result in the paper.
#11. A table containing the perfusion ratio values and standard deviations were added at the end of the paper.
#12. For the limited sample size, we did not do the correction for the multiple comparisons. It was indicated under the 'Statistics' section, page 6, last paragraph.